Author's response to reviews

Title: Prognostic significance of endothelial dysfunction in patients undergoing percutaneous coronary intervention in an era of drug eluting stent

Authors:

Motoki Kubo (k.moto.39@gmail.com)
Toru Miyoshi (miyoshit@cc.okayama-u.ac.jp)
Hiroki Oe (hirokioe@gmail.com)
Yuko Ohno (he17805@s.okadai.jp)
Kazufumi Nakamura (ichibun@cc.okayama-u.ac.jp)
Hiroshi Ito (itomd@md.okayama-u.ac.jp)

Version: 2  
Date: 7 May 2015

Author's response to reviews:

Timothy Shipley
Executive Editor
BMC Cardiovascular Disorders

Dear Dr. Shipley:

We are submitting our manuscript entitled “Prognostic significance of endothelial dysfunction in patients undergoing percutaneous coronary intervention in an era of drug eluting stent” for your consideration as a publication in BMC Cardiovascular Disorders. We wish this manuscript would be published in the category of Original Investigation.

All authors have read and approved the submission of the manuscript; the manuscript has not been published and is not being considered for publication elsewhere, in whole or in part, in any language, except as an abstract. There are no financial or other relations that could lead to a conflict of interest.

Thank you for your consideration.

Sincerely,

Toru Miyoshi
Department of Cardiovascular Medicine,
Okayama University Graduate School of Medicine, Dentistry and Pharmaceutical Sciences.